Author's response to reviews

Title: The antibacterial, phytochemicals and antioxidants evaluation of the root extracts of Hydnora africana Thunb. used as antidysenteric in Eastern Cape Province, South Africa.

Authors:

Olubunmi A Wintola (olubunmiwintola@yahoo.com)
Anthony J Afolayan (Aafolayan@ufh.ac.za)

Version: 4 Date: 22 August 2015

Author's response to reviews: see over
Answers to reviewer comments

Abstract

Page 1 line 49 "africna was changed to africana"

Results

Page 34 Table 2

"staphylococcus aureus was changed to Staphylococcus aureus"

Figure 1

"Phenol was changed to Phenolic" and "tanin to tannin"

Title of figure 1 was rephrased